Author's response to reviews

Title: A novel method for studying the temporal relationship between type 2 diabetes mellitus and cancer using the electronic medical record

Authors:

Adedayo A Onitilo (onitilo.adedayo@marshfieldclinic.org)
Rachel V Stankowski (stankowski.rachel@mcrf.mfldclin.edu)
Richard L Berg (berg.richard@mcrf.mfldclin.edu)
Jessica M Engel (engel.jessica@marshfieldclinic.org)
Gail M Williams (g.williams@sph.uq.edu.au)
Suhail A Doi (s.doi@sph.uq.edu.au)

Version: 2 Date: 23 July 2013

Author's response to reviews: see over
July 23, 2013

Joselito Acosta
Journal Editorial Office
BioMed Central
BMC Medical Research Methodology

MS: 9077846210323094 Research article
A novel method for studying the temporal relationship between type 2 diabetes mellitus and cancer using the electronic medical record
Adedayo A Onitilo, Rachel V Stankowski, Richard L Berg, Jessica M Engel, Gail M Williams and Suhail A Doi

Dear Ms. Acosta:

Thank you for considering our manuscript and giving us initial feedback. We are a bit confused by the comments. You first state that the paper is not suited for BMC Medical Research Methodology, but instead suggest BMC Medical Information and Decision Making. We are accepting of having our paper submitted to BMC Medical Information and Decision Making. However, your letter also asks us to revise the manuscript and upload the revision (see highlighted text in your original letter below.) We are following the instructions you have provided, but if you wish to transfer our manuscript to BMC Medical Information and Decision Making, please feel free to do so.

You ask us to clarify the authorization details for the health register data access and indicate whether we obtained approval from a data protection committee and institutional ethical review boards. **We have not applied for health register data access. Our manuscript does contain a statement that the study was approved by our institutional review board. Our institution does not utilize a separate data protection committee.**

All correspondence regarding this manuscript should be sent to Dr. Adedayo A. Onitilo, Marshfield Clinic Weston Center, 3501 Cranberry Boulevard, Weston, WI 54476. I can also be reached by telephone (715-393-1400), by fax (715-393-1399), or by email (onitilo.adedayo@marshfieldclinic.org).

Sincerely,

Adedayo A. Onitilo, MD, MSCR, FACP
Dear Dr. Onitilo,

Thank you for your recent submission of the manuscript detailed above.

We have now looked over your manuscript and we feel that it is out of scope for the journal. However, we do feel that the manuscript may be suited for another of our journals, **BMC Medical Informatics and Decision Making** (http://www.biomedcentral.com/bmcmedinformdecismak)

In addition, we would need you to address the following request below.

*******************************

**Further Requirements:**

(1) We would ask you to clarify the authorization details for the health register data access. Please clarify whether you obtained approval from a data protection committee and institutional ethical review boards. A statement to this effect must appear in the Methods section of the manuscript, including the name of the body which gave approval, with a reference number where appropriate.

*******************************

Once you have made these changes, please upload the revised version by following these instructions:

1. Go to ‘my BioMed Central’ (http://www.biomedcentral.com/my/) and log on with your email address and password. Then click on ‘My manuscripts’. You will reach a page giving details of all the manuscripts you have submitted.
2. Click on the ‘Submit revision’ button next to the title of this manuscript.
3. In the ‘Manuscript details’ tab, please update the title, abstract and author details if they have changed since the previous version.
4. In the ‘Cover letter’ tab, please provide a covering letter with a point-by-point description of the changes made.
5. In the ‘Upload files’ tab, please upload the revised version of the manuscript and press ‘Submit new version’. Please wait for the confirmation page to appear - this may take a few moments.

If you would be happy to have your manuscript transferred to this journal for further consideration, please let us know within the next seven days and we will handle the transfer on your behalf. Your manuscript will then be assessed by an Editor and sent for peer review if appropriate.

If we do not hear from you within the next seven days, we will have to proceed to close your file. If you wish to submit your manuscript to **BMC Medical Information and Decision Making** at any time later on, you can submit your files as a new manuscript via the URL provided above.

Please do not hesitate to contact us if you have any questions.

Best wishes,

Joselito

Joselito Acosta
Journal Editorial Office
BioMed Central

E: editorial@biomedcentral.com
W: www.biomedcentral.com